Case-finding and improving patient outcomes for chronic obstructive pulmonary disease in primary care: the BLISS research programme including cluster RCT

Peymané Adab,^{1*} Rachel E Jordan,^{1*} David Fitzmaurice,¹ Jon G Ayres,¹ KK Cheng,¹ Brendan G Cooper,² Amanda Daley,¹ Andrew Dickens,¹ Alexandra Enocson,¹ Sheila Greenfield,¹ Shamil Haroon,¹ Kate Jolly,¹ Sue Jowett,¹ Tosin Lambe,¹ James Martin,¹ Martin R Miller,¹ Kiran Rai,¹ Richard D Riley,³ Steve Sadhra,¹ Alice Sitch,¹ Stanley Siebert,⁴ Robert A Stockley⁵ and Alice Turner^{1,5}

¹Institute of Applied Health Research, University of Birmingham, Birmingham, UK ²Lung Function and Sleep, University Hospitals Birmingham NHS Foundation Trust, Birmingham, UK

- ³Centre for Prognosis Research, Research Institute for Primary Care and Health Sciences, Keele University, Keele, UK
- ⁴Business School, University of Birmingham, Birmingham, UK

⁵Respiratory Medicine, University Hospitals Birmingham NHS Foundation Trust, Birmingham, UK

*Corresponding author p.adab@bham.ac.uk and r.e.jordan@bham.ac.uk

Declared competing interests of authors: Peymané Adab reports grants from the National Institute for Health Research (NIHR) during the conduct of the study. She is chairperson of the NIHR Public Health Research (PHR) Funding Committee and a member of the NIHR PHR Prioritisation Committee. Rachel Jordan reports grants from NIHR during the conduct of the study. David Fitzmaurice reports membership of the NIHR Health Technology Assessment (HTA) Primary Care, Community and Preventative Interventions panel and membership of the NIHR HTA Prioritisation Committee. Kate Jolly reports grants from NIHR during the conduct of the study and is part funded by NIHR Collaboration for Leadership in Applied Health Research and Care West Midlands. She is a subpanel chairperson of the NIHR Programme Grants for Applied Health Research. Sue Jowett reports personal fees from work as an independent advisor at a Pfizer (Pfizer Inc., New York, NY, USA) chronic pain advisory board meeting in November 2018 outside the submitted work and is a member of the NIHR HTA Clinical Evaluation and Trials funding committee and the NIHR HTA programme Funding Board Policy Group. Richard Riley reports grants from NIHR HTA during the conduct of the study and personal fees from the BMJ outside the submitted work. Alice Sitch reports grants from NIHR during the conduct of the study and grants from AstraZeneca (AstraZeneca, Cambridge, UK) Investigator Initiated study outside the submitted work. Robert Stockley reports personal fees from AstraZeneca, Nycomed (Nycomed, Zurich, Switzerland), Boehringer Ingelheim (Boehringer Ingelheim, Ingelheim am Rhein, Germany), CSL Behring (CSL Behring, King of Prussia, PA, USA), Shire (Shire plc, Lexington, MA, USA), Chiesi (Chiesi Ltd, Manchester, UK), Polyphor (Polyphor Ltd, Allschwil, Switzerland), GlaxoSmithKline (GlaxoSmithKline plc, Brentford, UK), Mereo BioPharma (Mereo BioPharma Group plc, London, UK), Vertex Pharmaceuticals (Vertex Pharmaceuticals, Inc., Boston, MA, USA) and Akari Therapeutics (Akari Therapeutics plc, New York, NY, USA) outside the submitted work. Alice Turner reports grants from NIHR during the conduct of the study; grants, personal fees and non-financial support from AstraZeneca and Chiesi; personal fees and non-financial support from Pfizer outside the submitted work.

Published November 2021 DOI: 10.3310/pgfar09130

Plain English summary

BLISS research programme Programme Grants for Applied Research 2021; Vol. 9: No. 13 DOI: 10.3310/pgfar09130

NIHR Journals Library www.journalslibrary.nihr.ac.uk

Plain English summary

C hronic obstructive pulmonary disease is an important lung disease, affecting around 10% of adults worldwide. Each year in the UK, it accounts for 1.4 million general practitioner consultations, 1 million hospital bed-days and around 24 million lost working days. However, at least half of those with chronic obstructive pulmonary disease are unaware of their diagnosis and these people may not receive early treatment. Before our research there was uncertainty about (1) how to best identify these missing patients, (2) whether or not early identification would benefit patients, (3) how chronic obstructive pulmonary disease progresses, (4) what characteristics (other than smoking) affect risk of hospital admission or early death and (5) what aspects influence ability to work in people with chronic obstructive pulmonary disease.

We found that if general practitioners offered screening to smokers aged over 40 years then they could identify seven times as many people with chronic obstructive pulmonary disease than they do currently. However, although these patients could potentially benefit from therapies, the health system was not set up to support doctors to provide all the recommended treatments. Our economic model suggested that screening was worthwhile for detecting undiagnosed chronic obstructive pulmonary disease. However, after 4 years we found that screening did not reduce the risk of hospital admissions or death.

We also followed up around 2000 people with chronic obstructive pulmonary disease to see which features were linked with the risk of hospital admission with a lung problem. Through this we developed a tool that could measure an individual's chronic obstructive pulmonary disease severity. This has the potential to allow doctors to make more appropriate patient management decisions, but it needs more testing.

Finally, we examined which attributes (related to the patient, their lung problem or their workplace) affected people's ability to work. We found that people who are more breathless or exposed to inhaled hazards may have poorer work performance. However, because few patients in the study were in paid employment, we cannot draw firm conclusions.

© Queen's Printer and Controller of HMSO 2021. This work was produced by Adab *et al.* under the terms of a commissioning contract issued by the Secretary of State for Health and Social Care. This issue may be freely reproduced for the purposes of private research and study and extracts (or indeed, the full report) may be included in professional journals provided that suitable acknowledgement is made and the reproduction is not associated with any form of advertising. Applications for commercial reproduction should be addressed to: NIHR Journals Library, National Institute for Health Research, Evaluation, Trials and Studies Coordinating Centre, Alpha House, University of Southampton Science Park, Southampton SO16 7NS, UK.

Programme Grants for Applied Research

ISSN 2050-4322 (Print)

ISSN 2050-4330 (Online)

This journal is a member of and subscribes to the principles of the Committee on Publication Ethics (COPE) (www.publicationethics.org/).

Editorial contact: journals.library@nihr.ac.uk

The full PGfAR archive is freely available to view online at www.journalslibrary.nihr.ac.uk/pgfar. Print-on-demand copies can be purchased from the report pages of the NIHR Journals Library website: www.journalslibrary.nihr.ac.uk

Criteria for inclusion in the Programme Grants for Applied Research journal

Reports are published in *Programme Grants for Applied Research* (PGfAR) if (1) they have resulted from work for the PGfAR programme, and (2) they are of a sufficiently high scientific quality as assessed by the reviewers and editors.

Programme Grants for Applied Research programme

The Programme Grants for Applied Research (PGfAR) programme, part of the National Institute for Health Research (NIHR), was established in 2006 to fund collaborative, multidisciplinary programmes of applied research to solve health and social care challenges. Findings are expected to provide evidence that lead to clear and identifiable patient benefits, in the relatively near future.

PGfAR is researcher led and does not specify topics for research; however, the research must be in an area of priority or need for the NHS and the social care sector of the Department of Health and Social Care, with particular emphasis on health and social care areas that cause significant burden, where other research funders may not be focused, or where insufficient funding is available.

The programme is managed by the NIHR Central Commissioning Facility (CCF) with strategic input from the Programme Director. For more information about the PGfAR programme please visit the website: https://www.nihr.ac.uk/explore-nihr/funding-programmes/programme-grants-for-applied-research.htm

This report

The research reported in this issue of the journal was funded by PGfAR as project number RP-PG-0109-10061. The contractual start date was in January 2011. The final report began editorial review in February 2019 and was accepted for publication in February 2020. As the funder, the PGfAR programme agreed the research questions and study designs in advance with the investigators. The authors have been wholly responsible for all data collection, analysis and interpretation, and for writing up their work. The PGfAR editors and production house have tried to ensure the accuracy of the authors' report and would like to thank the reviewers for their constructive comments on the final report document. However, they do not accept liability for damages or losses arising from material published in this report.

This report presents independent research funded by the National Institute for Health Research (NIHR). The views and opinions expressed by authors in this publication are those of the authors and do not necessarily reflect those of the NHS, the NIHR, CCF, NETSCC, PGFAR or the Department of Health and Social Care. If there are verbatim quotations included in this publication the views and opinions expressed by the interviewees are those of the interviewees and do not necessarily reflect those of the authors, those of the NHS, the NIHR, NETSCC, the PGFAR programme or the Department of Health and Social Care.

© Queen's Printer and Controller of HMSO 2021. This work was produced by Adab *et al.* under the terms of a commissioning contract issued by the Secretary of State for Health and Social Care. This issue may be freely reproduced for the purposes of private research and study and extracts (or indeed, the full report) may be included in professional journals provided that suitable acknowledgement is made and the reproduction is not associated with any form of advertising. Applications for commercial reproduction should be addressed to: NIHR Journals Library, National Institute for Health Research, Evaluation, Trials and Studies Coordinating Centre, Alpha House, University of Southampton Science Park, Southampton SO16 7NS, UK.

Published by the NIHR Journals Library (www.journalslibrary.nihr.ac.uk), produced by Prepress Projects Ltd, Perth, Scotland (www.prepress-projects.co.uk).

NIHR Journals Library Editor-in-Chief

Professor Ken Stein Professor of Public Health, University of Exeter Medical School, UK

NIHR Journals Library Editors

Professor John Powell Chair of HTA and EME Editorial Board and Editor-in-Chief of HTA and EME journals. Consultant Clinical Adviser, National Institute for Health and Care Excellence (NICE), UK, and Professor of Digital Health Care, Nuffield Department of Primary Care Health Sciences, University of Oxford, UK

Professor Andrée Le May Chair of NIHR Journals Library Editorial Group (HS&DR, PGfAR, PHR journals) and Editor-in-Chief of HS&DR, PGfAR, PHR journals

Professor Matthias Beck Professor of Management, Cork University Business School, Department of Management and Marketing, University College Cork, Ireland

Dr Tessa Crilly Director, Crystal Blue Consulting Ltd, UK

Dr Eugenia Cronin Senior Scientific Advisor, Wessex Institute, UK

Dr Peter Davidson Consultant Advisor, Wessex Institute, University of Southampton, UK

Ms Tara Lamont Senior Scientific Adviser (Evidence Use), Wessex Institute, University of Southampton, UK

Dr Catriona McDaid Senior Research Fellow, York Trials Unit, Department of Health Sciences, University of York, UK

Professor William McGuire Professor of Child Health, Hull York Medical School, University of York, UK

Professor Geoffrey Meads Emeritus Professor of Wellbeing Research, University of Winchester, UK

Professor James Raftery Professor of Health Technology Assessment, Wessex Institute, Faculty of Medicine, University of Southampton, UK

Dr Rob Riemsma Reviews Manager, Kleijnen Systematic Reviews Ltd, UK

Professor Helen Roberts Professor of Child Health Research, UCL Great Ormond Street Institute of Child Health, UK

Professor Jonathan Ross Professor of Sexual Health and HIV, University Hospital Birmingham, UK

Professor Helen Snooks Professor of Health Services Research, Institute of Life Science, College of Medicine, Swansea University, UK

Professor Ken Stein Professor of Public Health, University of Exeter Medical School, UK

Professor Jim Thornton Professor of Obstetrics and Gynaecology, Faculty of Medicine and Health Sciences, University of Nottingham, UK

Please visit the website for a list of editors: www.journalslibrary.nihr.ac.uk/about/editors

Editorial contact: journals.library@nihr.ac.uk